Author's response to reviews

Title: Pregnant women's attitudes towards alcohol consumption

Authors:

Neil Raymond (mzyznr@nottingham.ac.uk)
Charlotte Beer (Charlotte.Beer@nottingham.ac.uk)
Cristine Glazebrook (cris.glazebrook@nottingham.ac.uk)
Kapil Sayal (Kapil.Sayal@nottingham.ac.uk)

Version: 3 Date: 2 June 2009

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

As outlined below, we have listed the amendments to the manuscript that you have requested. We look forward to hearing from you in due course.

Yours sincerely

Kapil Sayal

1) Reference 16 - Broken link. This has been updated and checked.

2) Email addresses – We have included email addresses for all the authors on the title page.

3) Table - Table has been cited as 'Table 1' rather than just 'Table'

4) Table vertical elements – These have been removed

5) Table placement text - This has been removed